Author’s response to reviews

Title: Evaluation of serum prolidase activity and oxidative stress markers in men with BPH and prostate cancer

Authors:

Faruk Kucukdurmaz (farukdr@hotmail.com)
erkan efe (drerkanefe@hotmail.com)
Ahmet Celik (drahmetcelik@gmail.com)
hasan dagli (hasandagli63@gmail.com)
metin kilinc (mkilinc@ksu.edu.tr)
sefa resim (sefaresim@gmail.com)

Version: 3 Date: 09 Nov 2017

Author’s response to reviews:

1- We reinstated the box plot version of the figure with a larger font size for the category labels on the x-axis. Also, we added more tick marks and labels on the y-axis and used a larger font size for these labels.

2- The missing spaces between consecutive words in the main text were also corrected.